Author's response to reviews

Title: The attractions of medicine: the generic motivations of medical school applicants in relation to demography, personality and achievement

Authors:

I C McManus (i.mcmanus@ucl.ac.uk)
G Livingston (g.livingston@ucl.ac.uk)
C Katona (c.katona@kent.ac.uk)

Version: 5 Date: 20 February 2006

Author's response to reviews:

Website addresses - Please cite website addresses in the reference list and remove them from the manuscript text. Cite them using the following format:

Website title [http://www.website.com].

I have done this and put the references in as 'Internet communication' in Reference Manager 11, with the web address in the Web/URL field. However the Biomedcentral output style provided on your website doesn't handle this properly (see references 35 - 37; please advise).